Author's response to reviews

Title: Using simple telehealth to manage hypertension in general practice: a service evaluation

Authors:

Elizabeth Cottrell (Elizabeth.Cottrell2@northstaffs.nhs.uk)
Tracey Cox (Tracey.Cox@stoke.nhs.uk)
Phil O'Connell (phil.oconnell@nhs.net)
Ruth Chambers (Ruth.Chambers@stoke.nhs.uk)

Version: 2
Date: 13 March 2015

Author's response to reviews:

Dear Editor

Many thanks for your email regarding our submitted manuscript. Please find attached the updated version in which we have:

1. Inserted line numbers
2. Changed the 'Funding' section to 'Acknowledgements' and provided greater detail about the source of funding of each of the authors.
3. Moved all the tables to after the references, keeping the captions in the position in the text to suggest the approximate area for insertion.

We hope that these changes are acceptable and we look forward to hearing back from you in due course.

Yours sincerely

Dr E Cottrell, on behalf of the authors